

Reuleaux, A and Siregar, BA and Collar, NJ and Panggur, MR and Mardiastuti, A and Jones, MJ and Marsden, SJ (2020) Protected by dragons: Density surface modeling confirms large population of the critically endangered Yellow-crested Cockatoo on Komodo Island. Condor: an international journal of avian biology, 122 (4). ISSN 0010-5422

Downloaded from: https://e-space.mmu.ac.uk/628885/

Version: Accepted Version

Publisher: Oxford University Press (OUP)

DOI: https://doi.org/10.1093/condor/duaa042

Please cite the published version

https://e-space.mmu.ac.uk

Protected by dragons: Density surface modeling confirms large population of the critically endangered Yellow-crested Cockatoo on Komodo Island

Anna Reuleaux,^{1,*} Benny A. Siregar,² Nigel J. Collar,³ Maria R. Panggur,⁴ Ani Mardiastuti,⁵ Martin J. Jones,¹ and Stuart J. Marsden¹

¹ Department of Natural Sciences, Manchester Metropolitan University, Manchester, UK

² Burung Indonesia, Bogor, Indonesia

³ BirdLife International, Cambridge, UK

⁴ Komodo National Park, Labuan Bajo, Indonesia

⁵ Bogor Agricultural University (IPB University), Bogor, Indonesia

*Corresponding author: anna.reuleaux@stu.mmu.ac.uk, anna.reuleaux@gmail.com

Submission Date: February 24, 2020; Editorial Acceptance Date: July 9, 2020; Published September 15, 2020

ABSTRACT

Intense trapping of the critically endangered Yellow-crested Cockatoo (*Cacatua sulphurea*) for the international pet trade has devastated its populations across Indonesia such that populations of >100 individuals remain at only a handful of sites. We combined distance sampling with density surface modeling (DSM) to predict local densities and estimate total population size for one of these areas, Komodo Island, part of Komodo National Park (KNP) in Indonesia. We modeled local density based on topography (topographic wetness index) and habitat types (percentage of palm savanna and deciduous monsoon forest). Our population estimate of 1,113 (95% CI: 587–2,109) individuals on Komodo Island was considerably larger than previous conservative estimates. Our density surface maps showed cockatoos to be absent over much of the island, but present at high densities in wooded valleys. Coincidence between our DSM and a set of independent cockatoo observations was high (93%). Standardized annual counts by KNP staff in selected areas of the island showed increases in cockatoo records from <400 in 2011 to ~650 in 2017. Taken together, our results indicate that KNP, alongside and indeed because of preserving its iconic Komodo dragons (*Varanus komodoensis*), is succeeding in protecting a significant population of Indonesia's rarest cockatoo species. To our knowledge this is the first time DSM has been applied to a critically endangered species. Our findings highlight the potential of DSM for locating abundance hotspots, identifying habitat associations, and estimating global population size in a range of threatened taxa, especially if independent datasets can be used to validate model predictions.

Keywords: Cacatua sulphurea, conservation, density surface model, distance sampling, habitat model, parrots, Psittaciformes, threatened species

LAY SUMMARY

- Yellow-crested Cockatoos are threatened by extinction due to illegal trapping for the pet trade. Komodo Island in Indonesia supports one of the largest remaining populations. The island is part of Komodo National Park, famous for its Komodo dragons. A 2006 survey indicated cockatoo numbers might have been declining.
- In contrast to previous surveys we sampled the whole island instead of focusing on coastal valleys, which are known to
 harbor the highest cockatoo densities. We used distance sampling and density surface modeling, which allowed us to
 estimate how many cockatoos remained undetected and to produce a map of predicted cockatoo densities.
- We estimate there are between 600 and 2,100 cockatoos on Komodo, most likely ~1,100. Komodo National Park authorities also reported an increase in their annual counts of cockatoos from below 400 in 2011 to around 650 in 2017. Thus, the cockatoo population on Komodo Island is large and stable; Komodo National Park is successfully protecting its cockatoos.

Terlindungi oleh komodo: density surface modeling pada Cacatua sulphurea (Kakatua-kecil Jambul-kuning) di Pulau Komodo

ABSTRAK (BAHASA INDONESIA)

Perburuan ilegal untuk perdagangan internasional terhadap burung *Cacatua sulphurea* (Kakatua-kecil Jambul-kuning) yang telah dikategorikan sebagai Kritis—telah menyebabkan penurunan populasi pada hampir semua lokasi di Indonesia, sehingga populasi dengan jumlah >100 individu hanya tersisa di beberapa tempat saja. Untuk menduga kepadatan dan

Copyright © American Ornithological Society 2020. All rights reserved. For permissions, e-mail: journals.permissions@oup.com.

jumlah populasi burung ini di Pulau Komodo (salah satu pulau di Taman Nasional Komodo; TNK), dilakukan penelitian dengan menggunakan perpaduan antara *distance sampling* dan *density surface modeling* (DSM). Permodelan kepadatan dilakukan berdasarkan topografi (*topographic wetness index*) dan tipe habitat. Dugaan populasi spesies burung ini di Pulau Komodo adalah 1.113 ekor (95% *Cl* 587–2.109), lebih tinggi dari pendugaan populasi sebelumnya. Peta kepadatan (*density surface maps*) menunjukkan bahwa kakatua ini tidak terdapat di sebagian besar pulau, namun dapat ditemukan dengan kepadatan yang tinggi di lembah-lembah berhutan. Kesesuaian antara DSM dan titik pengamatan independen bernilai tinggi (93%). Penghitungan populasi tahunan oleh staf Balai TNK pada lokasi-lokasi terpilih menunjukkan adanya penambahan populasi dari <400 ekor pada tahun 2011 menjadi sekitar 650 ekor pada tahun 2017. Dengan demikian, selain melindungi satwa komodo *Varanus komodoensis,* TNK juga berhasil melindungi populasi *Cacatua sulphurea* dalam jumlah yang signifikan. Penelitian ini merupakan upaya pertama yang menggunakan DSM untuk spesies dengan status Kritis. Metoda ini berpotensi untuk menentukan pusat-pusat kepadatan populasi, mengidentifikasi asosiasi habitat, serta menduga ukuran populasi secara global bagi taksa-taksa yang terancam punah, terutama jika dataset yang independen dapat dipakai untuk menvalidasi prediksi model.

Indonesian keywords (kata kunci): density surface model, distance sampling, kakatua, konservasi, model habitat, spesies terancam

INTRODUCTION

Estimates of population sizes are cornerstones of conservation science at both global and local scales and are instrumental in assessing extinction risks, conservation priorities, and Red List status (Mace et al. 2008, Collen et al. 2011). These essential data are, however, lacking for a great many rare and threatened species (MacKenzie et al. 2005), which are often difficult to survey on account of their biology and/or the areas they inhabit (McDonald 2004). Even for relatively well-known groups such as psittacines (parrots), ~75% of species are lacking abundance estimates (Marsden and Royle 2015), a worrying statistic given that almost onethird of psittacines are currently threatened (IUCN 2019). A variety of methods have been used to calculate population size in parrots. For very rare species it may be possible to count every individual. For others, marked or identifiable individuals allow mark-recapture or mark-resighting methods, but these conditions are not the norm. For most species, roost counts, flyway counts, and distance sampling have been used more or less effectively (Casagrande and Beissinger 1997, Marsden and Royle 2015). Distance sampling, despite difficulties in meeting method assumptions, has become a well-established method for estimating sizes of animal populations generally (Thomas et al. 2010) and parrots in particular (Marsden and Royle 2015).

Estimates derived from distance sampling have become the most commonly used, involving 84% of published parrot abundance estimates (Marsden and Royle 2015), despite question marks over reliability related to lack of records in rare species and idiosyncrasies of parrot behavior (Marsden 1999, Dénes et al. 2018). Alternative methods fail to measure absolute bird abundance (Bibby et al. 2000), face the same (and additional) challenges (Casagrande and Beissinger 1997), or remain largely untested (Dénes et al. 2018).

While there has been considerable work on optimizing distance sampling design, field protocol, and analysis

phases (Marsden 1999, Buckland 2006, Bächler and Liechti 2007, Margues et al. 2007, Buckland et al. 2008, Oedekoven et al. 2015), far less attention has been paid to the process of estimating site-based or total population sizes through extrapolation of local abundances at sampled sites to larger areas or even whole ranges of threatened birds. Several extrapolation methods have been used, including simple multiplication of average density by area of study site or range (e.g., Guix et al. 1999, Marques et al. 2007), stratification by habitat type (e.g., Jones et al. 1995, Casagrande and Beissinger 1997), and interpolation across unvisited sites (Koshkin et al. 2016). The best-accepted methods are those which model local density against habitat and other relevant features (Buckland et al. 2016), sometimes along with spatial information, to predict densities in unvisited areas (e.g., Williamson and Homes 1964, Somershoe et al. 2006). Apart from predicting spatial distributions and producing realistic abundance estimates, spatial modeling can also identify factors that affect abundance (Hedley and Buckland 2004), knowledge which can then inform conservation management decisions. The spatial input for the model can either originate from covariates with a spatial distribution (e.g., habitat, elevation, distance from coast) or include the location coordinates directly (usually latitude and longitude). The functional relationships between these covariates and the response variable are rarely linear in reality, and generalized additive models (GAMs) allow this to be reflected in complex nonlinear functions in the modeling process (Zuur et al. 2014).

Density surface modeling (DSM) uses GAMs (Wood 2017) to model the point-specific density at the sampling points (or segment-specific for line transects) in a 2-step approach: first, it accounts for detectability using the distance sampling method; second, it incorporates spatial and/or environmental covariates to explain the variation between sampling points (Hedley and Buckland 2004, Miller et al. 2013). The resultant model can then be used to map predicted population densities within the sampling

area and also, with caution, for new unsampled areas (Miller et al. 2013). DSMs are not widely used for population estimates at present but have been successfully applied to marine birds (Petersen et al. 2011, Winiarski et al. 2013, 2014; Bradbury et al. 2014), a peatland bird community (Leivits and Leivits 2016), marine mammals (de Segura et al. 2007, Gilles et al. 2011, Williams et al. 2011, Miller et al. 2013, Bravington et al. 2019), and ungulates (Harihar et al. 2014, Schroeder et al. 2014, La Morgia et al. 2015, Valente et al. 2016). Several of these studies had conservation objectives such as identification of priority areas for protection (Winiarski et al. 2013) or assessment of endangered species (Ibouroi et al. 2019). While the method has been recommended as suitable for parrots (Dénes et al. 2018), we know of no application of DSMs to any parrot, or indeed to any critically endangered species.

The Yellow-crested Cockatoo (Cacatua sulphurea) used to occur commonly across the Lesser Sunda Islands, parts of Sulawesi, and its satellites (Figure 1), but habitat alteration and especially excessive trapping for the international pet trade from the 1970s through the 1990s caused severe declines and local extinctions across much of its range. Thus, populations >100 individuals remain at only a handful of sites, rendering the species critically endangered (Broch 1981, Cahyadin et al. 1994a, 1994b; Jones et al. 1995, PHPA/LIPI/BirdLife International 1998, Agista et al. 2001, BirdLife International 2001, Eaton et al. 2015). Some 560-4,000 of the very distinctive subspecies citrinocristata are thought to exist in several forest patches on Sumba (Wungo 2011; A. Reuleaux personal observation; Figure 1). The population of Yellow-crested Cockatoos of the subspecies occidentalis in Komodo National Park (KNP; Figure 1), in the Lesser Sunda Islands, is also believed to be relatively large, although a survey in 2006 (Imansyah et al. 2016) diagnosed a sharp decline since 2000 (Agista and Rubyanto 2001). Both these surveys obtained minimum numbers for selected coastal valleys by direct sightings from vantage points, which cover <10% of the island's area. Here we used density surface modeling to predict local cockatoo densities across Komodo Island. We validated the models using independent sightings, investigated correlates of local abundance, and estimated the island-wide population size.

METHODS

Study Area

Komodo Island ($8.42^{\circ}S-8.75^{\circ}S$, $119.37^{\circ}E-119.57^{\circ}E$) is situated between Flores and Sumbawa in the Lesser Sunda Islands, Nusa Tenggara Timur, Indonesia (Figure 1). With an area of 340 km² and a maximum elevation of 824 m above sea level (m.a.s.l.), it is the largest and highest of the islands of KNP, which was established in 1980 to protect the Komodo dragon (*Varanus komodoensis*) and

the terrestrial and marine biodiversity of the islands (UNESCO World Heritage Committee 1991, Lilley 1997). It now harbors one of the most important remnant populations of Yellow-crested Cockatoos and the most important population of subspecies C. s. occidentalis (Collar and Marsden 2014). Komodo is situated in one of the driest areas of Indonesia; streams do not run for most of the year and natural water sources are rare (Monk et al. 1997). Large areas of the island are covered by open grassland (Auffenberg 1980) interspersed with scrubland, palm savanna, small stands of broadleaved trees, and gallery forests along watercourses (Monk et al. 1997). Where larger streams meet the sea, deciduous monsoon forests cover the valley floors (Auffenberg 1980, Monk et al. 1997; Figure 2). Higher altitudes (>500 m) support denser closed-canopy forest (Figure 2), which is often dominated by bamboo or rattan and referred to as "quasi cloud forest" (Auffenberg 1980) or "mossy forest" (Monk et al. 1997); this terrain transitions downhill via sparse forest into scrubland.

Following recommendations based on conservation considerations (Collar et al. 2017), to avoid supplying information to potential trappers, we do not include complete maps of our results. The complete maps are available for *bona fide* researchers or for conservation purposes from the authors. They are replaced in the results section by out-of-context cutouts of exemplary locations.

Point Count Distance Sampling

Komodo Island has a surface area of 340 km². We first excluded all 1×1 km² pixels that contained >50% bare grassland or sea (landcover map by Ministry of Environment and Forestry Indonesia, KLHK 2017), habitat types deemed unsuitable for cockatoos. From the remaining 152 potentially suitable pixels we randomly selected 25 for our point count distance sampling. The survey stations were located 200 m apart on the perimeter of each of these pixels (navigation by GPS). From November 6 to December 14, 2017, one of two experienced observers (AR, BAS) carried out one distance sampling point count at each of 178 stations between 0600 and 1000 hours (Figure 1). Both observers had experience in studying cockatoos (22 and 36 mo, respectively) and distance sampling of cockatoos (each 3 mo, early in 2017). The number of survey stations per pixel varied from 5 to 10 (mean = 7) depending on how many point counts could be finished within the survey time frame. Slow walking speed in rough terrain and large distances from the nearest permitted campsite often hindered maximization of survey effort.

We followed standard methods for point count distance sampling (Buckland et al. 2001, 2008; Thomas et al. 2010). Specifically we adapted the field protocol described by Marsden (1999): (1) 10-min count durations but without a settling-down period, (2) exclusion of encounters in flight,



FIGURE 1. Study area Komodo Island situated in Komodo National Park, Indonesia, showing 178 point count stations, nested within 25 sampling squares, with their 250-m radius buffers.

(3) recording of flocks as clusters including the number of their individuals, and (4) replacement of group sizes for purely aural detections with the average size of known groups. To minimize errors in assessing distances, we used laser rangefinders (Nikon Forestry Pro) and followed protocols suggested by Buckland (2006) and Buckland et al. (2008). For example, this included measuring distances to other objects at a similar distance if the detected bird was not directly visible. Our survey period fell in the early part of the breeding season (Agista and Rubyanto 2001). Although all pairs observed near cavities were still prospecting, we checked the surrounding of each survey station for cavities with incubating adults. Analysis followed standard methods recommended in Buckland et al. (2001) and used a truncation distance of 350 m and open vs. enclosed habitat as a 2-level covariate for the detection function. We defined stations as open habitat if palm

savanna, scrubland, and grassland made up $\geq 60\%$ of land cover within a 50-m radius). We carried out distance sampling analysis in R using package *Distance 1.0.0* (Miller et al. 2016, R Core Team 2019). We used ungrouped distances as recorded without manual binning. Cluster size as a covariate was very unstable against truncation distances and did not improve AIC, so no cluster size bias regression was used. Results are reported as means ± SE.

Environmental Variables

We used a wetness index (by System for Automated Geoscientific Analyses, SAGA), a topographic index predicting the soil moisture based solely on a digital elevation grid (Böhner and Selige 2006, Conrad et al. 2015), in our case a Shuttle Radar Topography Mission digital elevation model with \sim 30 × 30 m resolution. We generated a contemporary raster habitat map based on LANDSAT



FIGURE 2. Habitat types on Komodo Island: (A) gallery forest among open grassland; (B) gallery forest; (C) palm savanna in front of deciduous monsoon forest; (D) quasi cloud forest (mossy forest, >500 m.a.s.l.).

8 imagery from September and October 2015–2017 (Appendix Figure 4). After cloud removal and adjustment of burnt areas to the survey period, we used our field observations and Google Maps to generate training data for landcover classification in QGIS with the semi-automatic classification plug-in (Congedo 2016, QGIS Development Team 2019).

We tailored the classification for use as a predictor of cockatoo detectability and density, and distinguished 6 habitat types (Appendix Table 3) following Auffenberg (1980) and Monk et al. (1997): open grassland and scrubland; palm savanna; deciduous monsoon forest including gallery forests and monsoon forests of the coastal plains; mangrove forest; quasi cloud forest >500 m; and sparse forest as a transition zone between quasi cloud forest and open habitat types. After inspection of spectral signature plots of the training units, cutoff values for critical bands were set manually to improve separation of overlapping categories (Congedo 2016); grassland and scrubland were classified separately and pooled afterwards. Mangrove forest was not recognized properly by the classification method, so the Indonesian Ministry of Forestry's landcover data (KLHK 2017) were used instead to correct the extent of this locally rare habitat type. We used the resulting fineresolution habitat map to calculate percentage cover for each 250-m radius buffer around sampling stations and for each 0.25-km² prediction pixel. The 2 classes with the highest cockatoo encounter rates were termed "suitable habitat" and their combined percentages were used as a covariate in the DSM.

For model building we summarized the environmental covariates at the point count locations by averaging the gridded values within overlapping 250-m radius circles (sampling buffers) centered at each location. For our prediction surface, we divided each 1-km² pixel of the island into 4, resulting in 1,457 prediction grid pixels containing land. The environmental covariate values obtained at a smaller resolution were averaged within each of these 0.25-km² prediction pixels (Figure 3).

Density Surface Modeling (DSM) and Prediction

We used density surface modeling (Miller et al. 2013) to estimate population density within each 0.25-km² pixel across the island, involving the distance-based abundance estimates and the 2 environmental covariates. The *dsm* function in R (Miller et al. 2019) is based on generalized additive models (Wood 2017, 2019) and a detection function (Miller et al. 2016), and allows for the uncertainty of detection probability when estimating the variance of this 2-step modeling process. Our full generalized additive mixed model (GAMM) included the explicit spatial term s(x,y) (a smooth function for interaction of latitude and longitude), smooth functions of the log-transformed SAGA wetness index and the arcsine transformed percentage



FIGURE 3. Exemplary map details of predictor values (topographic wetness index and percentage of suitable habitat) and densities of Yellow-crested Cockatoos (individuals km⁻²) predicted by the density surface model, on a 0.25-km² grid of Komodo Island; cross-validated with independent sightings of the species (Agista and Rubyanto 2001, Imansyah et al. 2003, 2016; Taman Nasional Komodo 2016, eBird Basic Dataset 2019). To avoid supplying information to potential trappers, locations are provided out of context, with smoothed coastlines and random orientation. The complete maps of the whole island are available for research or conservation purposes from the authors.

of suitable habitat, as well as an autocorrelation structure (AR1 structure with form = ~ 1 |sampling square) to allow for nestedness of the point count stations within the sampling squares. For the spatial term, we used a Duchon spline (Duchon 1977) as recommended for areas with complex borders where misidentification of population

hotspots is a potential problem (Miller and Wood 2014). Smooth functions for interactions of these environmental variables were also explored but did not improve model fit. We compared Gaussian, Tweedie, and quasi Poisson distributions. After dropping nonsignificant terms, we selected the best combination of the remaining terms by AIC minimization (Appendix Table 4). Comparison of GAMMs is not straightforward; indicators are still in development (Wood 2017) and the AIC of the lme component of the GAMM is not recommended as an indicator for choice either (Wood 2019). Therefore, we used the AIC of the equivalent GAM for this step. With the GAM component of the chosen GAMM we predicted cockatoo density and coefficient of variation (CV) for each pixel of the prediction grid. To obtain the overall variance and confidence intervals we combined the variance of the detection function and that of the GAM using the Delta method via the dsm.var.gam function (Seber 1982, Miller et al. 2019).

Validating DSM Predictions Using Independent Cockatoo Sightings

We used 3 independent sources of cockatoo observations that were not included in our DSM analysis to validate the spatial predictions of our DSM. The first source was annual monitoring by KNP staff on flight paths and roosts (2012-2017, 16 locations; KNP unpublished data). The second source was citizen science observations from eBird (eBird Basic Dataset 2019), from which we selected those records where the observer had specified a precise location on the map instead of allocating it to a predefined hotspot, the national park, or the island in general (2004-2017, 7 locations). The third source was cockatoo records from survey reports, involving 9 locations from valleyfloor surveys in 2000 (Agista and Rubyanto 2001) and 10 locations from a general fauna survey in 2002 (Imansyah et al. 2003), which were partly confirmed again by cockatoo valley-floor surveys in 2005 and 2006 (Imansyah et al. 2005, 2016). We used only one independent sighting location per prediction pixel. We checked coincidence of the model's local density predictions against these known positives (regarding densities ≥ 1 individual km⁻² as predicted presence).

RESULTS

Cockatoo groups were observed at 48 of the 178 point count locations, with an encounter rate of 0.38 groups per point count (after exclusion of flying individuals and truncation). Encounter rates were highest in deciduous monsoon forest (0.91 ± 0.17, n = 22) and palm savanna (0.62 ± 0.11, n = 86) and lowest in the remaining habitat types (0.19 ± 0.10, n = 26 in grassland and scrubland; 0.06 ± 0.04, n = 32 in sparse forest; 0.00, n = 12 in quasi cloud forest; and no data in mangrove forest). The average number of individuals in each group seen was 2.61 individuals (SE = 0.49, n = 31, before truncation). Detection probabilities were described best by a hazard-rate detection function with habitat openness as a covariate (Table 1, Appendix Figure 5).

The DSM with the best fit contained 2 smooth terms with thin plate regression splines of 2 environmental covariates: SAGA wetness index (log-transformed, F = 8.08, edf = 1, P = 0.005; Appendix Table 4) and percentage cover by suitable habitat (palm savanna and deciduous monsoon forest combined and arcsine transformed, F = 7.70, edf = 1, P = 0.006; Figure 3, Appendix Figure 6). The spatial term was excluded as it was not significant (P > 0.3 regardless of spline base, as long as the model accounted for the autocorrelation structure of the points within sampling squares). This best model predicted high cockatoo densities (>8 individuals km⁻², locally up to 48 individuals km⁻²) for 2 forested valleys (Figure 3) where cockatoos are known to be common and where we had high encounter rates at point counts (2.77 \pm 0.49 individuals per station, *n* = 35, presence at 32, up to 6 groups at one station) and flock sizes up to 60 individuals in incidental observations. High densities were also predicted for a dry river valley and a coastal valley that we did not sample, along with moderate densities for several other unsampled locations (Figure 3). Crosschecking these locations with the independent sightings showed that the model had predicted almost all known cockatoo hotspots, and 93% of the 42 independent presence points. The 3 false negatives were very close (<85 m) to pixels with predicted cockatoo presence. The mapped coefficient of variation showed that CV was high in areas with predicted low densities and low in high-density areas. Totaling the modeled population densities over the whole

TABLE 1. Comparison of half-normal and hazard-rate detection functions with and without habitat openness as a covariate. Δ AlC between the 2 top models was small but visual inspection of the detection function confirmed the choice of a hazard-rate key detection function with openness as 2-level covariate, although resulting confidence intervals were slightly larger than with the equivalent halfnormal model. C-vM: Cramér-von Mises goodness-of-fit test. Δ AlC: difference in Akaike Information Criterion compared to best model.

Key function	Formula	df	C-vM P-value	Average detectability	SE (average detectability)	ΔAIC
Hazard-rate	~openness	3	0.98	0.247	0.050	0.0
Half-normal	~openness	2	0.77	0.224	0.030	1.0
Half-normal	~1	1	0.63	0.252	0.031	8.4
Hazard-rate	~1	2	0.76	0.248	0.055	9.0

island resulted in a population estimate for Komodo Island of 1,113 individuals (95% CI: 587–2,109; Figure 3).

DISCUSSION

We used density surface modeling (Miller et al. 2013) of local abundance estimates from distance sampling to estimate the population of the critically endangered Yellowcrested Cockatoo on the 340-km² island of Komodo. Our estimated population size of 1,113 individuals and the spatial density predictions are in line with independent KNP Authority monitoring, which recorded direct sightings of over 600 individuals (KLHK and DJ KSDAE 2018) when selectively covering <10% (albeit the most suitable areas) of the island. In their preference for palm savanna and deciduous monsoon forest Komodo's cockatoos resemble conspecifics on other islands, whereas their near-complete absence from quasi cloud forest is unexpected, as similar habitat types and altitudes are readily used on other islands (e.g., Jones et al. 1995, Trainor et al. 2008). While the absence from quasi cloud forest could be seasonal, there are no incidental sightings reported for other times of the year. Mangrove forest-an important roosting habitat for cockatoos on Komodo (Agista and Rubyanto 2001)-was not sampled because our survey times were deliberately chosen to avoid daily periods of high cockatoo mobility and commenced after the cockatoos had traveled away from their roosts early in the morning. Although the numbers are not directly comparable, we calculated a local abundance estimate for the pixels overlapping the valley-floor study areas used by Imansyah et al. (2016). For these valleys our model predicts a population size of 397 individuals, which is more than double the number of direct sightings in September and October 2005/2006 (Imansyah et al. 2016) but not far above the 340 individuals sighted in September-October 2000 in those areas (Agista and Rubyanto 2001).

Our results strongly suggest that the population on Komodo Island is substantial, and we found no evidence that the steep decline reported for the early 2000s has continued (Imansyah et al. 2016). Direct counts collected annually by experienced KNP rangers from vantage points overlooking 6 coastal valleys show an increase from <400 recorded cockatoos for Komodo Island in 2011 to 641 in 2017 (Table 2; Taman Nasional Komodo 2016; KNP unpublished data). This is evidence that the population has certainly been stable and probably increasing over the last 6 yr. As such, in addition to providing protection to the iconic Komodo dragon (Purwandana et al. 2014), KNP appears to be working as a long-term stronghold for the cockatoo. This park's population is by far the largest of the subspecies occidentalis and would be the largest for the entire species in the likely case that the distinctive C. s. citrinocristata is accorded species rank (Collar and Marsden 2014).

The remoteness and topography of Komodo Island and its fear-inducing dragon appear to provide some natural protection from habitat destruction (e.g., fires and conversion to agriculture) and illegal trapping, but enforcement of legal protection for the cockatoo by park authorities has undoubtedly played an important role in the current situation. Poor soils, steep terrain, and lack of water mean that there has never been much incentive for the single community on the island, which traditionally relied almost exclusively on fishing, to convert land for agriculture (Singleton et al. 2002, Pannell 2013). The Komodo dragons attract a stream of paying visitors (~180,000 in 2018; CNN Travel 2019) and, therefore, KNP is relatively well resourced (Hakim 2017, KLHK and DJ KSDAE 2018). It has 13 field stations, 120 staff (including a permanent presence of ~30 rangers on the islands), several speedboats, and provision for regular patrols and ecological monitoring (Taman Nasional Komodo 2016). Although patrols discover a few poaching incidents every year, these mainly concern marine life, and occasionally deer (2 cases of deer 2009-2015; Taman Nasional Komodo 2016). When over 40 young Komodo dragons were discovered in trade in 2019, they turned out to originate from the species' scarce populations outside KNP's borders (Gokkon 2019). The park has the support of local communities (Walpole and Goodwin 2001), which largely depend economically on tourism (Walpole and Goodwin 2000, 2001; Nurilma et al. 2019). Although KNP's fame, protection, visitors, income, and acceptance are mainly owed to Komodo dragons and marine life (UNESCO 1991), the cockatoos clearly benefit from the protection as well. KNP provides a successful model with regard to methods and resources that could be applied in other protected areas where formal protection has yet to increase cockatoo numbers.

Based on just 5 weeks of fieldwork, and despite the poor accessibility of most of the island and a complex mosaic of habitats, we succeeded in modeling the population of this difficult-to-count species with a distribution map that is suitable for conservation practitioners. Local cockatoo densities fall within the range of estimated densities of other cockatoo species (Marsden and Royle 2015) and the confidence intervals of the predicted densities are narrow enough to be used for assessing conservation status and viability. An independent dataset of cockatoo sightings gave us the opportunity to validate our predictions. During the modeling process this validation process in fact prevented us from accepting a candidate model that neglected residual spatial autocorrelation and instead included the spatial term as a predictor (Gaspard et al. 2019). This model, although favored according to information theoretic criteria, scored very poor hit rates on the independent sightings dataset, as it was dominated by the spatial term. This might be important for other researchers, as >80% of

TABLE 2. Minimum estimate of Yellow-crested Cockatoo numbers from annual monitoring by Komodo National Park authorities derived by summing direct encounters from simultaneous valley-floor counts (Taman Nasional Komodo 2016, KLHK and DJ KSDAE 2018; A. Kefi 2019 personal communication).

Island(s)	2011	2012	2013	2014	2015	2016	2017	2018	2019
Komodo	382	406	500	524	547	522	641	660	733
Rinca & Bero	111	136	149	122	148	160	141	151	150
Total	493	542	649	646	695	682	782	811	883

ecological and biogeographical modeling studies do not account for spatial autocorrelation (Gaspard et al. 2019), which can lead to estimation errors of coefficients of 25% on average (Dormann 2007). Accounting for autocorrelation was particularly important because of our clustered sampling design as opposed to studies where sampling locations are distributed more evenly. However, for parrot species and other rare, highly mobile birds in fragmented rugged habitats, sampling each inhabited patch will often be impractical.

As a 2-step modeling process, the DSM required that we combine the variances of both models (detection function and GAM) to obtain a realistic measure of the variance of our prediction. We used the Delta method (Seber 1982, Miller et al. 2019) for this purpose, ignoring a potential lack of independence between the 2 steps (stemming from the covariate in the detection function), because the more advanced variance propagation method (Williams et al. 2011) is not available for mixed models, and bootstrapping (Hedley and Buckland 2004) should not be used if smooth functions are involved in the model formula (Miller et al. 2013, Bravington et al. 2019). In general, we were forced by the combination of point transects, a covariate in the detection function, and an autocorrelation structure in the DSM to use mixed models (GAMMs instead of GAMs), for which more recent developments in the statistical software have yet to be incorporated. Optimizing adjustments (e.g., use of variance propagation or restricted maximum likelihood) might have increased the precision of our estimates, but spatial density predictions and estimated total population size were so stable across models and modeling engines that final results are unlikely to have differed.

Red List assessments of extinction risk currently rely heavily on population sizes and areas of occupancy (SSC IUCN 2001), but estimating these indicators for threatened species is often problematic as available resources limit precision and reliability of results. DSMs have the potential to provide these data based on limited sampling effort (La Morgia et al. 2015) because they cope well with nonrandom sampling designs (Miller et al. 2013) and can still predict absolute abundances and distributions (Hedley and Buckland 2004). They account for detection probability and utilize spatial environmental information, which is often available remotely. DSMs can also identify habitat associations or other ecological dependencies and predict population hotspots and range limits, which can be cross-validated with independent opportunistic datasets. However, despite their broad applicability for population estimates, DSMs have limitations as well: survey designs still need to cover the study area sufficiently (geographical extent, full range of densities including absence, all relevant habitat types and altitudes; Miller et al. 2013) and reach the minimum number of contacts for reliably estimating a detection function (Buckland et al. 2001). DSMs can only make useful predictions if the population's limiting factors can be captured directly or indirectly by spatially referenced covariates, and the method only reaches its full potential if these data are available remotely. Interpolated densities for unsampled areas between samples are predicted with confidence whereas extrapolation to new areas outside the sample range require more caution (Miller et al. 2013). Consequently, predictions across islands (or functional islands such as protected areas or areas that span biogeographical boundaries) that are not included at the modeling stage are risky, as new areas might be subject to unconsidered influences. In our case we decided against using the model from Komodo Island to estimate the neighboring cockatoo subpopulation on Rinca Island because additional factors such as introduced predators and accessibility for potential trappers from Flores could not be accounted for.

Cockatoos in KNP are a showcase for the potential of a 2-level monitoring approach, where annual trend assessment with relatively simple methods could be used to indicate optimal timing of high-effort abundance surveys like distance sampling with DSM. In the future large gaps in published abundance data, coinciding with suspected population declines as in the decades before our study (Imansyah et al. 2016), could be prevented if annual monitoring data are accessible to conservation practitioners who can then trigger more intensive research as soon as a decline becomes apparent and in time for potential mitigations. We found DSM to be an efficient and effective estimator of population size and distribution in the Yellow-crested Cockatoo, and suggest its use for larger populations of the species (e.g., on Sumba and in Timor-Leste), provided that region-specific limiting factors such as trapping pressure can be accounted for. The useful predictions and broad applicability of DSM give it an edge over alternative methods with similar survey effort and

make it a powerful tool for estimating population sizes of threatened island species.

ACKNOWLEDGMENTS

We thank our funders for their support; the staff of Komodo National Park (in particular A. Kefi, D. R. Hau, Sudarman, Mukhlas, W. Bubang, D. Ambu, Haruna, Ardiyanto, Safril, A. Latif, Y. B. Prakoso, N. Parera, H. Wong, F. I. Djoda, R. A. F. Putra, and A. Abdurahman) and Burung Indonesia (in particular T. Hani, F. Djawa, K. S. Rodriguez, L. Anjut, S. Rabenak, and D. Agista) for their help; the Indonesian Government for permission to undertake this research (Ristekdikti Research Permit 57/EXT/SIP/FRP/E5/Dit.KI/VII/2017 and KLHK/ KNP Simaksi SI.60/SET/HKST/Kumil/10/2017); our local counterparts Burung Indonesia, Bogor Agricultural University (IPB University), and Universitas Nusa Cendana Kupang for their support; Christian Devenish for advice on data analysis; and the referees and editors for their constructive input.

Funding statement: Our work on Komodo Island was funded by Zoologische Gesellschaft für Arten- und Populationsschutz (Fonds für bedrohte Papageien and Strunden Papageienstiftung). A.R. was supported by a scholarship of the German Academic Exchange Service (DAAD Doktorandenstipendium) and by Loro Parque Fundación. The funders did not have any input into the content of this manuscript, and did not require approval of the manuscript before submission or publication.

Ethics statement: This research was conducted in compliance with the ethics guidelines of Manchester Metropolitan University.

Author contributions: S.J.M. and N.J.C. conceived the idea, formulated the questions, and secured the funding; A.R. and B.A.S. developed the field protocol and collected the data; A.R. analyzed the data and wrote the manuscript; S.J.M., N.J.C., A.M., and M.J.J. supervised research and contributed to the manuscript; M.R.P. supervised and carried out data collection for the KNP annual counts dataset.

Conflict of interest statement: A.R., B.A.S., N.J.C., A.M., M.J.J., and S.J.M. declare that they have no conflicts of interest. M.R.P. declares that she is employed by Komodo National Park (KNP).

Data depository: Analyses reported in this article can be reproduced partly using data provided by Reuleaux et al. (2020). Sensitive location information was deleted from the dataset so an analysis cannot be reproduced completely with the data in Dryad. Uncensored data files are available from the authors to bona fide researchers.

LITERATURE CITED

- Agista, D., and D. Rubyanto (2001). Telaah awal status, penyebaran, dan populasi kakatua kecil jambul kuning, *Cacatua sulphurea parvula*: di Taman Nasional Komodo, Nusa Tenggara Timur. BirdLife International Indonesia Programme, Bogor, Indonesia.
- Agista, D., Sumardin, A. Hamid, F. N. Mallo, S. Alam, Harjun, and C. Mamengko (2001). Telaah awal status, penyebaran dan

Populasi Kakatua-kecil Jambul-kuning (*Cacatua sulphurea sulphurea*) di Taman Nasional Rawa Aopa Watumohai, Sulawesi Tenggara dan Pulau Pasoso, Sulawesi Tengah. BirdLife International - Indonesia Programme, Bogor, Indonesia.

- Auffenberg, W. (1980). The herpetofauna of Komodo with notes on adjacent areas. Bulletin of Florida State Museum, Biological Sciences 25:39–156.
- Bächler, E., and F. Liechti (2007). On the importance of g (0) for estimating bird population densities with standard distance sampling: Implications from a telemetry study and a literature review. Ibis 149:693–700.
- Bibby, C. J., N. D. Burgess, D. A. Hill, and S. Mustoe (2000). Bird Census Techniques, 2nd ed. Academic Press, Amsterdam, The Netherlands.
- BirdLife International (2001). Threatened Birds of Asia: The BirdLife International Red Data Book. BirdLife International, Cambridge, UK.
- Böhner, J., and T. Selige (2006). Spatial prediction of soil attributes using terrain analysis and climate regionalisation.
 In SAGA Analysis and Modelling Applications (J. Böhner, K. R. McCloy, and J. Strobl, Editors). Göttinger Geographische Abhandlungen, Göttingen, Germany. pp. 13–28.
- Bradbury, G., M. Trinder, B. Furness, A. N. Banks, R. W. G. Caldow, and D. Hume (2014). Mapping seabird sensitivity to offshore wind farms. PLOS One 9:e106366.
- Bravington, M. V., D. L. Miller, and S. L. Hedley (2019). Reliable variance propagation for spatial density surface models. arXiv:1807.07996v4 [stat.ME] https://arxiv.org/pdf/1807.07996
- Broch, H. B. (1981). Cultural variation on the islands in the Sea of Flores. Archipel 22:43–53.
- Buckland, S. T. (2006). Point-transect surveys for songbirds: Robust methodologies. The Auk 123:345–357.
- Buckland, S. T., D. R. Anderson, K. P. Burnham, J. L. Laake, D. L. Borchers, and L. Thomas (2001). Introduction to Distance Sampling: Estimating Abundance of Biological Populations. Oxford University Press, Oxford, UK.
- Buckland, S. T., S. J. Marsden, and R. E. Green (2008). Estimating bird abundance: Making methods work. Bird Conservation International 18:91–108.
- Buckland, S. T., C. S. Oedekoven, and D. L. Borchers (2016). Modelbased distance sampling. Journal of Agricultural, Biological, and Environmental Statistics 21:58–75.
- Cahyadin, Y., P. Jepson, and S. Arif (1994a). The status of *Cacatua sulphurea abbotti* on the Masalembo Islands. PHPA/BirdLife International Indonesia Programme, Bogor, Indonesia.
- Cahyadin, Y., P. Jepson, and M. Syarief (1994b). A rapid status assessment of *Cacatua sulphurea sulphurea* in South Sulawesi Province. PHPA/BirdLife International - Indonesia Programme, Bogor, Indonesia.
- Casagrande, D. G., and S. R. Beissinger (1997). Evaluation of four methods for estimating parrot population size. The Condor 99:445–457.
- CNN Travel (2019). Indonesia cancels plan to ban tourists from Komodo Island. CNN Travel, October 2. https://edition.cnn. com/travel/article/indonesia-komodo-islands-reverse-touristban-intl-hnk/index.html
- Collar, N. J., and S. J. Marsden (2014). The subspecies of Yellowcrested Cockatoo *Cacatua sulphurea*. Forktail 30:23–27.
- Collar, N., J. Eaton, and B. Sykes (2017) Conservation and the redaction of locality data. BirdingASIA 28:3–4.
- Collen, B., L. McRae, S. Deinet, A. De Palma, T. Carranza, N. Cooper, J. Loh, and J. E. M. Baillie (2011). Predicting how populations

decline to extinction. Philosophical Transactions of the Royal Society B 366:2577–2586.

Congedo, L. (2016). Semi-automatic classification plugin documentation. http://dx.doi.org/10.13140/RG.2.2.29474.02242/1

- Conrad, O., B. Bechtel, M. Bock, H. Dietrich, E. Fischer, L. Gerlitz, J. Wehberg, V. Wichmann, and J. Böhner (2015). System for Automated Geoscientific Analyses (SAGA) v. 2.1.4. Geoscientific Model Development 8:1991–2007.
- Dénes, F. V., J. L. Tella, and S. R. Beissinger (2018). Revisiting methods for estimating parrot abundance and population size. Emu-Austral Ornithology 118:67–79.
- de Segura, A. G., P. S. Hammond, A. Cañadas, and J. A. Raga (2007). Comparing cetacean abundance estimates derived from spatial models and design-based line transect methods. Marine Ecology Progress Series 329:289–299.
- Dormann, C. F. (2007). Effects of incorporating spatial autocorrelation into the analysis of species distribution data. Global Ecology and Biogeography 16:129–138.
- Duchon, J. (1977). Splines minimizing rotation-invariant seminorms in Sobolev spaces. In Constructive Theory of Functions of Several Variables (W. Schempp and K. Zeller, Editors). Springer, Berlin, Germany. pp. 85–100.
- Eaton, J. A., C. R. Shepherd, F. E. Rheindt, J. B. C. Harris, S. B. van Balen, D. S. Wilcove, and N. J. Collar (2015). Trade-driven extinctions and near-extinctions of avian taxa in Sundaic Indonesia. Forktail 31:1–12.
- eBird Basic Dataset (2019). Version: EBD_relMay-2019. Cornell Lab of Ornithology, Ithaca, NY, USA. https://ebird.org/science/ download-ebird-data-products
- Gaspard, G., D. Kim, and Y. Chun (2019). Residual spatial autocorrelation in macroecological and biogeographical modeling: A review. Journal of Ecology and Environment 43:Article 19.
- Gilles, A., S. Adler, K. Kaschner, M. Scheidat, and U. Siebert (2011). Modelling harbour porpoise seasonal density as a function of the German Bight environment: Implications for management. Endangered Species Research 14:157–169.
- Gokkon, B. (2019). Indonesia arrests 7 for allegedly selling Komodo dragons over Facebook. Mongabay, April 5. https:// news.mongabay.com/2019/04/
- Guix, J. C., M. Martín, and S. Mañosa (1999). Conservation status of parrot populations in an Atlantic rainforest area of southeastern Brazil. Biodiversity and Conservation 8:1079–1088.
- Hakim, L. (2017). Managing biodiversity for a competitive ecotourism industry in tropical developing countries: New opportunities in biological fields. AIP Conference Proceedings 1908(1):030008. American Institute of Physics, Malang, Indonesia.
- Harihar, A., B. Pandav, and D. C. MacMillan (2014). Identifying realistic recovery targets and conservation actions for tigers in a human-dominated landscape using spatially explicit densities of wild prey and their determinants. Diversity and Distributions 20:567–578.
- Hedley, S. L., and S. T. Buckland (2004). Spatial models for line transect sampling. Journal of Agricultural, Biological, and Environmental Statistics 9:181–199.
- Ibouroi, M. T., N. A. Hassane, S. Moindjié, M. Ombade, N. Mohamed, M. H. Saidou, K. Abderemane, A. Cheha, and J. Chiffard (2019). The first comprehensive survey of habitat suitability and population size for the endangered Grande Comoro Scops Owl (*Otus pauliani*): Implications for its conservation. Journal of Ornithology 160:1121–1132.

- Imansyah, M. J., D. G. Anggoro, N. Yangpatra, A. Hidayat, and Y. J. Benu (2005). Sebaran dan karakteristik pohon sarang kakatua jambul kuning (*Cacatua sulphurea parvula*) di pulau Komodo, Taman Nasional Komodo. CRES Komodo Project, TN Komodo, Denpasar, Indonesia.
- Imansyah, M. J., D. Purwandana, A. Ariefiandy, Y. J. Benu, T. S. Jessop, and C. R. Trainor (2016). Valley-floor censuses of the critically endangered Yellow-crested Cockatoo *Cacatua sulphurea occidentalis* on Komodo Island, East Nusa Tenggara province, Indonesia, point to a steep population decline over a six-year period. Forktail 32:66–72.
- Imansyah, M. J., D. Purwandana, H. Rudiharto, and T. S. Jessop (2003). Survey potensi hidupan liar terestrial di Pulau Komodo, Taman Nasional Komodo 2002. Zoological Society of San Diego, Taman Nasional Komodo, The Nature Conservancy, Labuan Bajo, Indonesia.
- IUCN (2019). The IUCN Red List of Threatened Species, version 2019-2. http://www.iucnredlist.org
- Jones, M. J., M. D. Linsley, and S. J. Marsden (1995). Population sizes, status and habitat associations of the restricted-range bird species of Sumba, Indonesia. Bird Conservation International 5:21–52.
- KLHK (2017). Penutupan Lahan Indonesia. Landcover map Indonesia by WebGIS Ministry of Environment and Forestry. http://webgis.menlhk.go.id:8080/kemenhut
- KLHK and DJ KSDAE (2018). Statistik Direktorat Jenderal Konservasi Sumber Daya Alam dan Ekosistem Tahun 2017. Kementerian Lingkungan Hidup dan Kehutanan, Jakarta, Indonesia.
- Koshkin, M. A., R. J. Burnside, N. J. Collar, J. L. Guilherme, D. A. Showler, and P. M. Dolman (2016). Effects of habitat and land use on breeding season density of male Asian Houbara *Chlamydotis macqueenii*. Journal of Ornithology 157:811–823.
- La Morgia, V., R. Calmanti, A. Calabrese, and S. Focardi (2015). Costeffective nocturnal distance sampling for landscape monitoring of ungulate populations. European Journal of Wildlife Research 61:285–298.
- Leivits, A., and M. Leivits (2016). Using density surface modelling for estimating abundance of breeding birds in heterogeneous mire habitats: A case study in Estonia. In BirdNumbers 2016: Birds in a Changing World. Programme and Abstracts of the 20th Conference of the European Bird Census Council (Dachverband Deutscher Avifaunisten, Editor). Halle, Germany.
- Lilley, R. J. (1997). Komodo National Park. In The Ecology of Nusa Tenggara and Maluku. (K. Monk, Y. De Fretes and G. Reksodiharjo-Lilley, Editors). eBook. Periplus Editions, Hong Kong. pp. 807–811.
- Mace, G. M., N. J. Collar, K. J. Gaston, C. Hilton-Taylor, H. R. Akçakaya, N. Leader-Williams, E. J. Milner-Gulland, and S. N. Stuart (2008). Quantification of extinction risk: IUCN's system for classifying threatened species. Conservation Biology 22:1424–1442.
- MacKenzie, D. I., J. D. Nichols, N. Sutton, K. Kawanishi, and L. L. Bailey (2005). Improving inferences in population studies of rare species that are detected imperfectly. Ecology 86:1101–1113.
- Marques, T. A., L. Thomas, S. G. Fancy, and S. T. Buckland (2007). Improving estimates of bird density using multiple-covariate distance sampling. The Auk 124:1229–1243.
- Marsden, S. J. (1999). Estimation of parrot and hornbill densities using a point count distance sampling method. Ibis 141:327–390.
- Marsden, S. J., and K. Royle (2015). Abundance and abundance change in the world's parrots. Ibis 157:219–229.

- McDonald, L. L. (2004). Sampling rare populations. In Sampling Rare or Elusive Species: Concepts, Designs, and Techniques for Estimating Population Parameters (W. L. Thompson, Editor). Island Press, Washington, DC, USA. pp. 11–42.
- Miller, D. L., and S. N. Wood (2014). Finite area smoothing with generalized distance splines. Environmental and Ecological Statistics 21:715–731.
- Miller, D. L., M. L. Burt, E. A. Rexstad, and L. Thomas (2013). Spatial models for distance sampling data: Recent developments and future directions. Methods in Ecology and Evolution 4:1001–1010.
- Miller, D. L., E. Rexstad, L. Burt, M. V. Bravington, and S. Hedley (2019). dsm: Density Surface Modelling of Distance Sampling Data. https://cran.r-project.org/package=dsm
- Miller, D. L., E. Rexstad, L. Thomas, L. Marshall, and J. L. Laake (2016). Distance sampling in R. BioRxiv:63891.
- Monk, K., Y. De Fretes, and G. Reksodiharjo-Lilley (1997). Ecology of Nusa Tenggara and Maluku, 1st ed. Oxford University Press, Oxford, UK.
- Nurilma, A. D., A. Sjarmidi, and Y. Yustiana (2019). Economic valuation of Komodo dragon (*Varanus komodoensis*, Ouwens 1912) and its contribution to locally-generated revenue of West Manggarai Regency, East Nusa Tenggara Province. In IOP Conference Series: Earth and Environmental Science, Bandung Indonesia.
- Oedekoven, C. S., J. L. Laake, and H. J. Skaug (2015). Distance sampling with a random scale detection function. Environmental and Ecological Statistics 22:725–737.
- Pannell, S. (2013). Nature and culture in a global context: A case study from World Heritage Listed Komodo National Park, Eastern Indonesia. Terra Australis 36:53–63.
- Petersen, I. K., M. L. MacKenzie, E. Rexstad, M. S. Wisz, and A. D. Fox (2011). Comparing pre- and post-construction distributions of Long-tailed Ducks *Clangula hyemalis* in and around the Nysted Offshore Wind Farm, Denmark : A quasi-designed experiment accounting for imperfect detection, local surface features and autocorrelation. University of St Andrews, St Andrews, UK.
- PHPA/LIPI/BirdLife International (1998). Species Recovery Plan Yellow-crested Cockatoo. PHPA/LIPI/BirdLife International -Indonesia Programme, Bogor, Indonesia.
- Purwandana, D., A. Ariefiandy, M. J. Imansyah, H. Rudiharto, A. Seno, C. Ciofi, D. A. Fordham, and T. S. Jessop (2014). Demographic status of Komodo dragons populations in Komodo National Park. Biological Conservation 171:29–35.
- QGIS Development Team (2019). QGIS geographic information system. Open Source Geospatial Foundation Project, Zanzibar, Tanzania. http://qgis.org
- R Core Team (2019). R: A Language and Environment for Statistical Computing. R Foundation for Statistical Computing, Vienna, Austria. http://www.r-project.org/
- Reuleaux, A., B. A. Siregar, N. J. Collar, M. R. Panggur, A. Mardiastuti, M. J. Jones, and S. J. Marsden (2020). Data from: Protected by dragons: Density surface modeling confirms large population of the critically endangered Yellow-crested Cockatoo (*Cacatua sulphurea*) on Komodo Island. The Condor: Ornithological Applications 122:1–15. doi:10.5061/dryad.47d7wm3b9
- Schroeder, N. M., S. D. Matteucci, P. G. Moreno, P. Gregorio, R. Ovejero, P. Taraborelli, and P. D. Carmanchahi (2014). Spatial and seasonal dynamic of abundance and distribution of guanaco and livestock: Insights from using density surface and null models. PLOS One 9:e85960.

- Seber, G. A. F. (1982). The Estimation of Animal Abundance and Related Parameters, 2nd ed. McMillan Publishing Company, New York, NY, USA.
- Singleton, J., R. Sulaiman, and TNC staff (2002). Environmental assessment study – Komodo National Park Indonesia. The Nature Conservancy-SE Asia Center for Marine Protected Areas, Bali, Indonesia.
- Somershoe, S. G., D. J. Twedt, and B. Reid (2006). Combining Breeding Bird Survey and distance sampling to estimate density of migrant and breeding birds. The Condor 108:691–699.
- SSC IUCN (2001). IUCN Red List categories and criteria: version 3.1. https://portals.iucn.org/library/node/10315
- Taman Nasional Komodo (2016). Statistik Balai Taman Nasional Komodo tahun 2009–2015. Kementerian Lingkungan Hidup dan Kehutanan, Labuan-Bajo, Indonesia.
- Thomas, L., S. T. Buckland, E. A. Rexstad, J. L. Laake, S. Strindberg, S. L. Hedley, J. R. B. Bishop, T. A. Marques, and K. P. Burnham (2010). Distance software: Design and analysis of distance sampling surveys for estimating population size. Journal of Applied Ecology 47:5–14.
- Trainor, C. R., F. Santana, P. Pinto, A. F. Xavier, R. Safford, and R. Grimmett (2008). Birds, birding and conservation in Timor-Leste. Birding Asia 9:16–45.
- UNESCO World Heritage Committee (1991). Convention concerning the protection of the World Cultural and Natural Heritage. https://whc.unesco.org/en/list/609/
- Valente, A. M., T. A. Marques, C. Fonseca, and R. T. Torres (2016). A new insight for monitoring ungulates: Density surface modelling of roe deer in a Mediterranean habitat. European Journal of Wildlife Research 62:577–587.
- Walpole, M. J., and H. J. Goodwin (2000). Local economic impacts of dragon tourism in Indonesia. Annals of Tourism Research 27:559–576.
- Walpole, M. J., and H. J. Goodwin (2001). Local attitudes towards conservation and tourism around Komodo National Park, Indonesia. Environmental Conservation 28:160–166.
- Williams, R., S. L. Hedley, T. A. Branch, M. V. Bravington, A. N. Zerbini, and K. P. Findlay (2011). Chilean blue whales as a case study to illustrate methods to estimate abundance and evaluate conservation status of rare species. Conservation Biology 25:526–535.
- Williamson, K., and R. C. Homes (1964). Methods and preliminary results of the Common Birds Census, 1962–63. Bird Study 11:240–256.
- Winiarski, K. J., D. L. Miller, P. W. C. Paton, and S. R. McWilliams (2013). Spatially explicit model of wintering Common Loons: Conservation implications. Marine Ecology Progress Series 492:273–283.
- Winiarski, K. J., D. L. Miller, P. W. C. Paton, and S. R. McWilliams (2014). A spatial conservation prioritization approach for protecting marine birds given proposed offshore wind energy development. Biological Conservation 169:79–88.
- Wood, S. N. (2017). Generalized Additive Models: An Introduction with R, 2nd ed. Chapman and Hall/CRC, London, UK.
- Wood, S. N. (2019). mgcv: Mixed GAM computation vehicle with automatic smoothness estimation. R-project, https://cran.rproject.org/package=mgcv
- Wungo, E. (2011). Laporan survei populasi paruh bengkok terancam punah *Cacatua sulphurea citrinocristata*, Sumba, Nusa Tenggara Timur. Burung Indonesia, Bogor, Indonesia.
- Zuur, A., E. N. leno, and A. A. Saveliev (2014). A Beginner's Guide to Generalized Additive Mixed Models with R. Highland Statistics Ltd., Newburgh, UK.



APPENDIX FIGURE 4. Komodo Island habitat classification derived from supervised classification of LANDSAT 8 satellite images with training data from Google Maps. Forest types follow Auffenberg (1980) and Monk et al. (1997) with addition of sparse forest as a transitional zone between quasi cloud forest and open habitat types. Mangrove forest was not distinguished in the supervised classification and was added afterwards (KLHK 2017).



APPENDIX FIGURE 5. Hazard-rate key detection function (**A**) and detection probability density function (**B**) with openness of the habitat as 2-level covariate. Open circles represent individual detections and show the influence of the covariate: in open habitats detection probabilities remained high farther away from the observer (points above the line in [**A**], slopes shallower than the line in [**B**]) than in other habitats (points below the line in [**A**], slopes steeper than the line in [**B**]).



APPENDIX FIGURE 6. Shape of the smooth functions used as environmental predictors for Yellow-crested Cockatoo densities in the GAMM. "Wetness" is the log-transformed SAGA wetness index and "habitat" is the percentage cover by suitable habitat (palm savanna and deciduous monsoon forest combined and arcsine transformed). Ticks on the *x*-axis indicate the sample distribution.

Habitat class (this study)	Class in Auffenberg 1980	Class in Monk et al. 1997	Location	Main characteristics	Cockatoo observations at point counts ^a	Cover on Komodo Island
Open grassland & scrubland	Steppe	Savanna	Mostly lowlands	Treeless	6	51.5%
Palm savanna	Savanna forest	Savanna	Large lowland areas and many small fragments	Open with tall <i>Borassus</i> and <i>Corypha</i> palms	37	18.6%
Deciduous monsoon forest	Deciduous monsoon forest	Gallery forest, dry monsoon forest, moist deciduous monsoon forest	Along rivers and in coastal valleys	Closed canopy, <i>Tamarindus</i> , <i>Sterculia</i> , and <i>Bredelia</i>	25	16.5%
Mangrove forest	Mangrove forest	Mangrove forest	Tidal zone	Mangrove species	0	0.5%
Quasi cloud forest	Quasi cloud forest	Mossy forest	>500 m	Moss and lichen on trees, bamboo, rattan	0	3.7%
Sparse forest	Transitional zone to quasi cloud forest	Not mentioned	<500 m, transition zone from quasi cloud forest to scrubland	No closed canopy, bamboo groves	0	9.2%

APPENDIX TABLE 3.	Habitat type classification of	n Komodo Island modified from .	Auffenberg (1980) and Monke	et al. (1997).

^a Number of encounters in our distance sampling survey during point counts.

APPENDIX TABLE 4. Model choice for the density surface model. In model names **S** stands for spatial term (a smoother to the Duchon spline base of the interaction of geographic coordinates s(x,y)), **W** for wetness index (log transformed, s(wetness)), and **H** for the cumulative percentage cover by suitable habitat types (with an arcsine transformation, s(habitat)); edf = effective degrees of freedom, AIC(GAM) = AIC of the equivalent Generalized Additive Model.

Predictors	edf	CI	Р	AIC(GAM)
W + H ª				468
(Intercept)	1.25	0.55-1.95	0.001	
s(wetness)	1		0.005	
s(habitat)	1		0.006	
S + W + H				468
(Intercept)	1.27	0.58-1.96	<0.001	
s(x,y)	0		0.445	
s(wetness)	1		0.003	
s(habitat)	1		0.005	
W				473
(Intercept)	1.6	1.03-2.17	<0.001	
s(wetness)	1		<0.001	
н				474
(Intercept)	1.44	0.75-2.13	<0.001	
s(habitat)	1		<0.001	

^a Chosen model.